TWP Ops Pacific Teleconference Action Items 090ct07

Local Times:

USA: 16:30 MDT Tuesday Afternoon Direct dial-in numbers:

Darwin: 08:00 Wednesday Morning 1-303-433-3431 (international)
PNG: 08:30 Wednesday Morning 1-877-407-0183 (toll free in U.S.)
Nauru: 10:30 Wednesday Morning Access number: 329169

PARTICIPANTS:

TWPO ARCS#1 ARCS#2 ARCS#3

Larry Jones Hymson Waffi Andrew Kaierua Troy Culgan
Kim Nitschke Taddy Baras Franklin Teimitsi John Glowacki
Amon—Haruta Myra Selan Mike Ryczek
Chartes Brinkmann

Monty Apple

*Red Text indicates new info since last call*
**Blue Text indicates info on items discussed during call**
Tasks are organized alphabetically for each site.

1. Announcements
e Troy and Mike in Manus for SV0710M - Cal Visit. Both depart Manus
today.

® Kasis Inape of the PNG NWS in Port Moresby will be traveling to Manus
on 080ct, departing on 120ct.

2. Upcoming Site Visits (no need to cover during Teleconference)
® Krzysztof Krzton is in Heselbach until 11Nov07.

¢ SVO711M: 15-20Nov (emergency trip): Eagan - BOM tech accompany VSAT
technician to install new modem

e SVO0711N: 23Nov-07Dec07. Krzton + Glowacki. Cal Visit.

3. Manus Facility Status:
Discuss during Teleconference:

® BBSS: 0Old Digicora II in stores should be shipped back to SGP. It is
packed and ready to go.

® GENSET: Did Hastings Deering Fitter arrive and perform Genset Service?
No. Monty will follow up with HD as to why Fitter did not travel to
Manus during SV0710M.

® GENSET: Monty to find out if 750cca batteries will suffice or if
1000cca batts will need to be installed.

® HVAC: Status of purchasing 3 window ac units. Waiting on update from
Tania. Larry will follow up with Tania.

® Suominet: Troy to review email from Mentor about metpak. Manus
Suominet web page looks normal.

Issues to monitor, no need to cover on Teleconference:

e BBSS: GC-25 s/n 724434 - sent to SGP for upgrade or replacement on
01Aug07. Chris Keenan is sending 2 CD’s of the upgrade to SGP (1 for
Manus and 1 for Nauru). SGP will send a spare GC-25 to Manus when they
get a replacement or 724434 upgraded.

e ES&H: Conduit that is screwed down to concrete is %” diam x 30 feet
long (runs from power box by the E-Van to the Y-Van). On next SV,



evaluate if cable can be run overhead as opposed to location in
conduit on ground.

® Genset: Maintenance is due. Monty requested Hasting Deering to send
Tech during SVO0710M.

e HVAC: Y-van and 2, ISS ac unit’s to be replaced. AC units available in
Lorengau. Hymson will work with Dick Pearse to purchase. Moleon Sion
to help Hymson with installation and replacement.

® Security & Vans: Re-locate the current security lights pole inward so
that it stands next to the X-Van and new outside roof. Hymson to
proceed with construction. Hymson contacted the contractor to do the
work. No update as to when the contractor will start the work. Kasis
will purchase light in Port Moresby and send to Electrician so he can
install.

® VANS: I-van floor soft spots in marine plywood under linoleum.
BOM/Darwin to determine inspection/repair/monitor procedure.

® VANS: Outside roof construction: Contractor needs $ up front to
purchase materials. Tania/Larry to address with Dick Pearse.

Shipments: All exemptions are in - just waiting on equipment to arrive.

e BBSS, Cal, LGR, MMCR, MWR: Shipment SGP, 9/14/07, FedEx#
859558278109/Qantas 081-3599-9283, BBSS balloons, MMCR HD's & cases,
MWR blower, Fibers, BBSS temp sensors, ARM CorePC DVD, LGR modules,
Cal sensors, etc to Manus, 4 boxes

® GENSET: Shipment; Cargo Transport System, 11Sep07, Gazelle Coast Voy
191, S00015583 & 00019458, Ladder, 0il, Batteries, Refrigerant, to
Manus, 9 boxes

e MMCR: Shipment: ORNL, 12Sep07, DHL 8809470342, MMCR HD’s 1 & 2, to
Manus, 1 Case

® ORG: Shipment Darwin, 010ct07, DHL 342 5015 020, ORG Test Unit (s/n
0048), to Manus, 1 box.

® PWR: Shipment; Cargo Transport System in Brisbane, Gazelle Coast Voy
190 LCL, BoL S00015070, 6 ea 12V Gell Cell Batteries, to Manus, 6
boxes. Hymson will follow up with Pearse on this shipment.

4. Nauru Facility Status:
Discuss during Teleconference:
e COMMS: VSAT, DS1, and R1 status. VSAT bolts rusting badly - need
replacing. R1 back up. Monty to follow up on ordering bolts for both
Manus and Nauru.
® Genset: Diesel Fuel situation: Tank 1 (600 gallon tank) is 7/8 full.
Tank 2 (1000 gallon tank) is 7/8 full. 90 full drums remaining.
® Observer’s PC: Not functioning correctly. John building a GX620 and
installing software for the Observer’s PC.

® Pick-up truck: 0il change and servicing was due at 46,970km, vehicle
now at 52,618km. Please organize service. Vehicle serviced last week -
Andrew will forward current odometer reading.

® POWER: Need to change out 2 older 12V batteries in instrument field
power supply with 2 new batteries just received. Changed out Thursday
040ct07.

Issues to monitor, no need to cover on Teleconference:

e ES&H: Need new emergency lights for D, E, & I vans. Monty purchased.
BOM to send new first aid kit.

e IRT: Please verify the cable is plugged in at both ends for the irt
computer. Andrew checked connections. Waiting on response from Vic
Morris - as to adjustments required. Baud rate fixed and is ok now.



Our Airline: Currently only operating 1 flight per week into Nauru -
on Monday from Brisbane, and departing on Tuesday. Andrew states that
only one flight per week, Fridays arriving at 0430 and departing 1100
for Australia.

VEHICLE: Andrew will obtain quote to have truck tow hitch made and
installed on Nauru. Andrew sent quote to Tania. Larry will follow up
with Tania and get back to Andrew.

Shipments: Capt La Perouse Voy 26: ETD Brisbane 130ct07, ETA Nauru 230ct07.

BBSS: Shipment SGP, 7/12/07, FedEx# 8595 5827 8050/Qantas 081-3576-
8401, 120- BBSS radiosondes & 120- BBSS balloons to Nauru, 5 boxes.
Genset: Shipment, Cargo Transport System (CTS), 15Sep07, Captaine La
Perouse Voy 25 S00014516, 50 drums diesel fuel, to Nauru. Rec’d
020ct07.

OTHER: Shipment CTS (Cargo Transport System), 15Sep07, Captaine La
Perouse Voy 25 S00015578, Ladder, Refrigerant, 2 AC units, Genset 0il,
Batteries, to Nauru, 10 pcs. Rec’d 020ct07.

5. Darwin Status:

Discuss during Teleconference:

AERI: Blower fan needs dedicated 240V circuit to keep fan from
tripping and requiring reset during high outside temperatures. KK to
source local 240V fan and follow up with Denny Hackel on rewiring to
accomplish task.

COMMS: John working on software documentation and updates for TWP
sites.

Shipments: DHL has put a block on the account for shipments (Mae Chu’s
account) from Darwin and Germany. LANL account number is being used
for time being until Mae'’s is straightened out.

SMET: John to send status of Anemometer re-cabling with connectors and
lightning modules for TWP and AMF.

Issues to monitor, no need to cover on Teleconference:

Ceilometer: Need at least 1 good spare compensation fiber at Darwin.
Need to follow up with SGP/Vaisala on the 3 fibers that were sent from
Darwin and arrived SGP on 23Mar07. Vaisala is to test and send good
fibers to SGP for shipment to Darwin. Currently there are no spare
compensation fibers at Darwin.

DISD: Krzysztof to build splash panel similar to the one at SGP. This
is the dry season now so this is not a priority.

HVAC: HVAC training for BOM Techs. This will not take place till after
21Jan08.

MMCR: Labview errors. Kevin W. to address on next visit.

VANS: D-Van floor repair continues.

Shipments:

Loggers: Shipment Darwin, 12Sept 07,DHL 162 7267 224, Loggers to cal
(5959, 5961, 5967, 5971) 1 box. Rec’d at Campbell Australia on
19Sep07.

MMCR: Shipment Darwin, 27 Sept 07,DHL 162 7267 202, MMCR HD's to ORNL,
2 boxes.

MPL, SMET: Shipment Darwin, 28 sept 07, DHL 343 2089 636, MPL
detector, T/RH probes, power injector, to SGP, 1 box.

OTHER: Ship: LANL, 22Aug07, DHL7577517511, Meeting CD, to Darwin, 1
envelope

6. AMF Status - Germany Deployment: Ops 02Apr06 - 31December 2007.




Discuss during Teleconference:

AERI: Hatch issues being worked. AERI to be moved to AOS Van and Hatch
replaced during November visit.

CIMEL: Still not tracking properly.

MPL: per DQPR 1794, starting on 27Apr07: Vertical stripes are showing
up in the MPLPOL backscatter plots. Temp fluctuation problem.

MWRHF: DQPR 1813 issued on 05Sep07: RH is consistently 20-30% lower
than co-located MET and MWRP RH.

RWP: DQPR 1788: RWP stopped getting signals. Likely reason is failure
of the final amplifier. System was turned off on Aug 3, 2007 and amp
sent to SGP for forwarding to Vaisala. SGP sent to Vaisala on 19Sep07.
WACR: Faulty Antenna drier removed from service. Compressor motor is
running continuously despite display saying that the compressor is
off. Motor is running very hot and occasionally drops out when it
overheats. Replacement Drier ordered on 010ct07 by Tania, rec’d at SGP
on 020ct07.

Issues to monitor, no need to cover on Teleconference:

AERI: This version of AERI Hatch had wiring problems and are being
addressed by SSEC. BCR-01399 has been opened.

CIMEL: CIMEL is not tracking the sun properly. Instrument alignes with
the sun when GOSUN routine run, but moves off the sun when TRACK
routine is run.

Genset: Basler file: Shunt trip disconnected.

Genset: Need to replace alternator. Spare on site as well as spare
battery charger.

Genset: There is no 12V at the pick up point in the control box which
should be powering the auxiliary fuel pump. The auxiliary fuel pump
and racor fuel filter system have been bypassed as this pump was not
getting 12 volts to operate properly and the main inline pump can not
pull fuel through the racor filters by itself. The aux 12 volt pump is
not needed if the racor filter system is bypassed since the external
tank and the internal tank gravity equalize. Inline fuel filter should
be changed every 2 months.

Shipments:

CIMEL: Ship: NASA Goddard, 14Aug07, Fed X 9177 4738 1889/Lufthansa
020-3422-2322, CIMEL wet sensor, to AMF Germany, 1 envelope.

RWP: Shipment SGP, 9/19/07, Fed Ex# 791766728318, 1 - RF Power
Amplifier, 1 - Power Supply Unit, 1 - Modulator, IF & Interface, RWP
Computer, To Vaisala, 3 Boxes.

WACR: Shipment: AMF, 21Sept07, DHL 552 6516 332, WACR disk #3 to ORNL,
1 box.

WACR: Shipment: AMF, 28Sept07, DHL 552 6516 170, WACR disk 4 to ORNL,
1 box.

VANS: Shipment, Tom Muirhead (T&S Locating), 050ct07, Truck/Container
Ship/Truck, Balloons & Sondes, AOS Tower, and Misc, to AMF Germany, in
20" cargo container

7. AMF Status - China Deployment:

8. SGP — Instrument repair/cal status

AERI: Shipment SGP, 9/12/07, Fed Ex# 792560227733, 1 - GPS Receiver, 1
- GPS Antenna (from Darwin), To Univ. of Wisconsin, 1 Box

BBSS: Receival SGP, 9/13/07, UPS# 17215792V6643489901, 6 - GC25
Temperature Sensors, From Vaisala, 1 Box



BBSS, SMET: Shipment SGP, 9/4/07, Fed Ex# 791380002160, 1 - Ground
Check Set (z24434 from Manus), 2- Barometers (P0830007 & A0420001 from
AMF), 3 - TRH Sensors (V3630004, 21930018, & A1210034 from AMF), To
Vaisala, 1 Box

BBSS: Receival SGP, 9/24/07, UPS# 170116950349815898, 1 - Ground Check
Set (Z24434) (M), From Vaisala, 1 Box

CEIL: Shipment Darwin, 15 Aug 07, DHL 343 2089 662, Ceilometer (s/n
R0850006), TSI mirror, Logger (s/n 5964), to SGP , 2 boxes. Ceil & 2
Optical Fibers to Vaisala (Non tropical unit and receiver is slowly
dieing). CR23X logger return for checking, reference RMA# 4298. TSI
mirror to Vic Morris. AERI GPS Antenna and Receiver - ship to U of
Wisc. Rec’d SGP 23Aug07.

CEIL: Shipment SGP, 9/10/07, Fed Ex# 799711394334, 1 - Ceilometer
(RO850006), 2 - Optical Fibers, Ship to Vaisala, 1 Box.

CIMEL: Receival SGP, 8/9/07, Fed EX# 790799449059, 1 - Cimel Sys. BNL
#168, 1 - GPS Antenna, 1- V2TH Antenna, 1 - Satlink Logger, 1 -
Moisture Sensor, From AMF Germany, 1 box. Send to Wayne Newcomb at
NASA for repair and Calibration.

Data Logger: Shipment SGP, 8/8/07, Fed Ex# 790309924910, 1 -
Micrologger CR23X (D) s/n 7116, To Campbell Scientific for repair, 1
Box. Rec’d at Campbell on 10Aug07. Receival SGP, 9/6/07, UPS#
1728414970369534892, 1- Data Logger (7116 Darwin), From Campbell
Scientific, 1 Box.

Data Logger: Shipment SGP, 9/4/07, Fed Ex# 791380004656, 3 - Data
Loggers (6116, 7830, & 5964), To Campbell Scientific, 1 Box.

Data System: Receival SGP, 9/17/07, Fed Ex# 753355574264, 8

Lightning Arrestors, From Technology Integration Group, for Germany
(send 2 to Eagan for testing), 1 Box.

ECOR: Shipment SGP, 8/31/07, Fed Ex# 791379080963, 1 - LI-COR Gas
Analyzer, To LI-COR for evaluation at LICOR and Calibration at SGP
(from Germany), 1 Box. Rec’d back at SGP for calibration on 27Sep07.
IRT: Shipment SGP, 5/4/07, Fed Ex # 799634589832, 2 — IRT's (863 from
Darwin & 869 from Nauru), To Wintronics Inc., 1 box

IRT: Receival SGP, 9/17/07, Fed Ex# 753355574264, 4 - IRT's {863 (D),
866 (N), 867 (N), 868 (N)}, From Wintronics, 1 Box

MFRSR: Receival SGP, 12/28/06, DHL# 359 6718 182, 1- MFRSR logger
board (3DB2) & 1- MFRSR sensor head (00437) from AMF Niger, 1 box -
send to PNNL for Post Cal and then to Germany. Shipment SGP, 1/30/07,
FedEx# 7990 7836 6004, 1- MFRSR logger board & 1- MFRSR sensor head to
PNNL- Schmelzer (AMF-N), 1 box

MMCR: Shipment Manus, 0l1Aug07, DHL AWBR # 827 9131 965, TWTA, Coherent
Converter, Net RS, IRT Mirror, Ground Check Set, Temp Sensor, 16 Port
Switch, To SGP Via Global Logistics, 2 Boxes. Shipment SGP, 9/4/07,
Fed Ex# 799707088279, 1 - Coherent Up & Down Converter, To Spacek Labs
Inc., 1 Box. Per K. Widener on 19Sep07: Spacek following up on
receiver instability and a PLO was found not functioning properly.
MMCR: Receival SGP, 8/21/07, Fed Ex Freight# 791743538687, 1 — TWT
Amplifier (WD30130), From Manus, 1 Box. Shipment SGP, 9/5/07, Fed Ex
Freight# 791380404635, 1 - Traveling Wave Tube Amplifier, 1 - TWTube,
To Applied Systems Engineering, 1 Crate

ORG: Shipment SGP, 3/20/07, FedEx# 7912 5752 9906, 1- ORG (Model 115-
DA, s/n 03327-C7) to Optical Scientific (M), 1 box

Radiometers: Shipment Darwin, 14Sep07, DHL 3432089651, Radiometers to
SGP, 1 Box.



Next

RWP: Shipment SGP, 9/19/07, Fed Ex# 791766728318, 1 — RF Power
Amplifier, 1 - Power Supply Unit, 1 - Modulator, IF & Interface, RWP
Computer, To Vaisala, 3 Boxes.

SMET: Receival SGP, 9/4/07, UPS# 1Z7A03920469362574, 4 New - TRH Meter
Indicatiors, 4- Probes, From Vaisala, 1 Box

SMET: Shipment SGP, 8/8/07, Fed Ex# 792536018577, 9 - Humidity/Temp.
Indicators & Probes, To Vaisala for cal or repair, 1 Box. Rec’d at
Vaisala on 10AugQO7. Received back at SGP on 8/30/07.

SMET: Shipment SGP, 9/10/07, Fed Ex# 790332767375, 1 - Anemometer &
Propeller (AMF 60332), 2 - Anemometer Drives (from Darwin), To R.M.
Young, 1 Box

SMET: Receival SGP, 9/7/07, Fed Ex# 726931586642, 1 - Temp—-Hum Sensor
(V3630003) for Darwin, From ANL-Ritsche, 1 Box V3630003

SMET: Receival SGP, 9/24/07, UPS# 1Z247W0160442591256 , 2 - Barometers,
(s/n A0420001) (G) and (P0830007) (G), From Vaisala, 1 Box

SMET: Receival SGP, 9/25/07, UPS# 1Z0116950350054083, 3 - Temp-
Humidity Probes (s/n V3630004, 71930018, and A1210034) From Vaisala, 1
Box (originally from Germany) .

TSI: Shipment SGP, 5/1/07, FedEx# 7907 2912 4679, 1- TSI control
interface board to BNL for repair - Edwards, 1 box

Teleconference Call on 160ct07



